

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hornung, Erik

Working Paper Immigration and the Diffusion of Technology: The Huguenot Diaspora in Prussia

ifo Working Paper, No. 114

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Hornung, Erik (2011) : Immigration and the Diffusion of Technology: The Huguenot Diaspora in Prussia, ifo Working Paper, No. 114, ifo Institute - Leibniz Institute for Economic Research at the University of Munich, Munich

This Version is available at: https://hdl.handle.net/10419/73817

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Working Papers

Immigration and the Diffusion of Technology: The Huguenot Diaspora in Prussia

Erik Hornung

Ifo Working Paper No. 114

November 2011

An electronic version of the paper may be downloaded from the Ifo website www.cesifo-group.de.

Immigration and the Diffusion of Technology: The Huguenot Diaspora in Prussia*

Abstract

This paper analyzes long-term effects of skilled-worker immigration on productivity for the Huguenots migration to Prussia. We combine Huguenot immigration lists from 1700 with Prussian firm-level data on the value of inputs and outputs in 1802 in a unique data base. In 1685, religious persecution drove highly skilled Huguenots out of France into backward Brandenburg-Prussia where they were channeled into towns to compensate population losses due to plagues during the Thirty Years' War. Exploiting this settlement pattern in an instrumental-variable approach, we still find causal effects of Huguenot settlement on the productivity of textile manufactories hundred years after their immigration.

JEL Classification: N33, J24, O33, F22.

Keywords: Migration, technological diffusion, human capital, Huguenots, Prussian economic history

Erik Hornung Ifo Institute Poschingerstr. 5 81679 Munich, Germany Phone: +49(0)89/9224-1272 hornung@ifo.de

^{*} Comments from Sascha O. Becker, Davide Cantoni, Francesco Cinnirella, William Collins, Oliver Falck, Guido Schwerdt, Ludger Woessmann and seminar participants at Ifo Munich and the annual meetings of the European Society for Population Economics 2010, the German Economic Association 2010, the Spring Meeting of Young Economists 2010 and the Economic History Association 2010 are gratefully acknowledged. This work was financially supported by the Pact for Research and Innovation of the Leibniz Association and an Economic History Association Exploratory Travel and Data Grant 2009. Andrea Hohenadl provided capable research assistance.

1 Introduction

The impact of immigration on the host economy is a controversial issue both in research and in the political debate. Gains from skilled immigration are generally argued to be large if the productive abilities of immigrants are different from those of natives (Borjas, 1995). The imported human capital can either have direct effects on the productivity of a country, or external effects on the productivity of native human capital through the interpersonal transfer of technology and skill (Bodvarsson and Van den Berg, 2009). While the literature provides evidence for immediate productivity gains from immigration (Peri, 2011), empirical evidence for the indirect effects on the productivity of natives, which might occur only in the long run, is largely missing. With the help of historical data, this paper aims to analyze such long-term effects of skilled-worker immigration on productivity in the host country.

We construct a unique new dataset combining historical sources that document the immigration of French Protestants into Prussia and the productivity of Prussian manufactories roughly 100 years after their immigration. The religious flight of the so-called Huguenots in the seventeenth century is one of the earliest examples of skilled mass migration leading to technological transfers (Scoville, 1952a,b). Using Prussian immigration lists from 1700 that document the settlement of Huguenots precisely, we are able to observe the immigration of a population that was, on average, more skilled than the native population. The data are unique in the sense that Prussia was the only host country to keep exact records of the French refugees. In combination with firm-level data on the value of inputs and outputs for all 693 textile manufactories in 1802, we find positive long-term effects of immigration on productivity. To our knowledge, both data sources have not been used in previous econometric analysis.

We argue that, by the order of centralized ruling by the King and his agents, Huguenots were channeled into Prussian towns in order to compensate for severe population losses during the Thirty Years' War (1618-1648). This allows us to interpret the settlement pattern as a natural experiment and to use the population losses in an instrumentalvariable approach. Connecting data on the population decrease in Prussian towns during the war with the share of Huguenots in Prussian towns, we are able to identify exogenous variation in the settlement pattern and eliminate a possible selectivity of choosing the location of settlement for immigrants. Using this IV approach, we find manufactories established in towns that were depopulated due to disease and famine during the war, and subsequently repopulated by the immigration of Huguenots, achieve a higher productivity in manufacturing textiles than others. Population losses during the war were arguably exogenous to economic conditions of a town. However, to overcome possible concerns of non-random placement we provide several falsification and validation tests corroborate the validity of our IV approach.

The economic impact of the Huguenots, who fled their country after the Edict of Nantes was revoked in 1685, has been a recurring theme in the literature. As early as in the middle of the nineteenth century, List (1856, p. 153) found that "Germany owes her first progress in manufactures to the revocation of the Edict of Nantes, and to the numerous refugees driven by that insane measure into almost every part of Germany...". We interpret this event as a exogenous shock, after which immigrants fled from religious persecution and settled in the predominantly Protestant neighbor countries of France.

The consequences of this exogenous shock are not well studied. Using anecdotal evidence Scoville (1952a) denies immediate returns from Huguenot immigration. In the short run, their arrival neither accelerated economic growth in England, nor closed the technological gap that separated Germany from France, Holland, or England. Nevertheless, Scoville (1960, p. 363) argues that the cost of accommodating the Huguenots was easily offset by the long-term gains. While contemporary and present-day literature provides cost-benefit analysis of Huguenot immigration, effects from knowledge spill-overs between refugees and natives which might only be observed in the long run are neglected. It is well known that the Huguenots were highly trained and skilled, and that on arrival at their destinations they started to use these superior skills to earn a living (Scoville, 1953). Most recently Fourie and von Fintel (2011) found that immigrant Huguenots in Dutch South Africa preserved their comparative advantage in wine production over the eighteenth century. In other cases, interaction with native workers might have led to a transfer of technical knowledge and to technological diffusion. Accordingly, German scholars agree on the fact that the transfer of Huguenot knowledge had a certain positive effect on the Prussian economy (Jersch-Wenzel, 1978; Mittenzwei, 1987; Wilke, 1988b), econometric evidence is however missing.

The analysis of historical data has certain advantages since we are not only able to analyze long-term effects of immigration, but also to eliminate alternative channels of knowledge transfer. Technology features a tacit element, which requires direct communication between the user and the instructor of a new technology (Mokyr, 2002). As the growing number of modern forms of communication facilitate direct communication, measuring the unbiased effects from immigration is increasingly challenging. We avoid such problems by analyzing migration at a time when direct communication was virtually the only way of transferring technological knowledge. We exploit the fact that any means of indirect communication are negligible in this period and that immigrants in 1700 cannot have had any direct influence on productivity in 1802. Therefore, any productivity gain from immigration during this time is most likely to have been caused by interpersonal as well as intergenerational transfers of technology and skill.

The remainder of the paper is structured as follows. Section 2 gives insight into the related migration literature. Section 3 provides the historical background of Huguenot immigration into Brandenburg-Prussia. Section 4 introduces the dataset and its sources. In section 5 we formulate the empirical model and introduce the instrumental-variable approach. Section 6 presents OLS and IV results and tests their robustness. Section 7 concludes.

2 Economic Effects of Migration

The economic effects of modern migration are well-documented in the literature, especially in labor economics. This field distinguishes three major streams of research: the economic performance of immigrants, their effect on employment opportunities and wages of the natives, and the assessment of immigration policies for host countries (surveys by Borjas (1994, 1999); Friedberg and Hunt (1995)). The effect of immigration on natives' wages and labor market responses are certainly the most discussed, whereas the macroeconomic effects of immigration lack attention in the literature (Drinkwater et al., 2007). Studies of economic benefits from migration for the host country include Ben-Gad (2004), Chiswick et al. (1992) and Paserman (2008). The impact of immigration on innovation has been studied by Chellaraj et al. (2008), Hunt and Gauthier-Loiselle (2010) and Niebuhr (2010). An overview of historical migration and its impact is provided by Hatton (2010).

Theoretically, Borjas (1994, p. 1667) finds that immigrants with high levels of productivity who adapt rapidly to the labor market in the host country can make a significant contribution to economic growth. Borjas' (1995) influential "immigration surplus" finds immigration beneficial in a case where the immigrant skills are very different from those of the natives and their characteristics have a certain complementary to the native factors of production. He also finds that the knowledge transfer between natives and immigrants generates external effects leading to increasing returns to scale. In a simple model including capital he finds that the benefits from migration are large if the immigrants are skilled, thus having higher complementary with capital. This is even more relevant when the native population is rather unskilled. Borjas' findings are supported by Dolado et al. (1994) who find that immigration with low human capital is equal to an increase of the population - it slows down per capita growth. Similarly, if immigrants carry high levels of human capital which is complementary to native capital, per capita growth accelerates.

The unbiased analysis of knowledge transfers from immigrants to natives is almost impossible since diffusion processes are often affected by indirect channels of communication, like written or electronic media. Nevertheless, even if techniques are codified and explicit, interpretation by the user is needed. Successful interpretation thus requires a transfer of tacit knowledge between the instructor and the user (Mokyr, 2002). This means that even in today's environment, measuring the effect of transferring tacit knowledge through immigration is of particular interest to the understanding of technological diffusion.

A time when face-to-face contact was the only way to transfer knowledge provides the perfect setting to analyze the unbiased effects from technological diffusion through migration. Before the onset of the Industrial Revolution, innovation and diffusion rarely occurred as a result of the publication of written material or blueprints (Rosenberg, 1970), but rather through the migration of skilled craftsmen, financiers and entrepreneurs (Schilling, 1983, p. 8). At that time, the strongest obstacles to technological diffusion were mobility costs (Epstein, 2004). Furthermore, Cipolla (1972) notes that the effects of the printed word on historical diffusion of innovations are often overestimated, and direct communication was much more important when it came to application.

During the sixteenth and seventeenth centuries, mercantile policies started to gain influence over manufacturing and were aimed at stimulating innovation. Mokyr (1990, p. 78) provides some vivid examples of technological diffusion encouraged by European rulers. They attracted skilled foreign labor in order to apply foreign skills in the new host country and eventually transfer it to the natives. The literature widely agrees that this was a common way to diffuse knowledge during the Early Modern Ages and that host countries benefited substantially (Ciriacono, 2005; Findlay, 1978). Furthermore, it is agreed that Calvinists contributed substantially to the diffusion of technological knowledge during that time. The most famous example of Calvinist-migration was the exodus of Huguenots from France to the German Brandenburg-Prussia. Religious persecution increased the benefits from migration and thus helped to overcome obstacles to technological diffusion. In line with the aforementioned considerations of Borjas, Scoville (1951) argues that diffusion of skills and technologies was facilitated by the fact that Germany was a backward country in 1685.

One important caveat prevalent in the migration literature is that immigrant inflow is rarely accidental and immigration policies are most likely to be highly selective in the attraction of certain characteristics. Furthermore, it is often argued that immigrants are more mobile than natives and will move to regions with higher wages and probability for economic success. Usually this leads to two kinds of selection: selection based on the characteristics of the immigrants and selection based on their places of settlement. In our case we prefer selection on the characteristics of the immigrants since we presume that they were highly skilled on average. In order to analyze benefits from the knowledge transfer without observing individual skills of immigrants, we assume that the immigrants are pre-selected and, on average, more skilled than the natives. The second kind, selection on the place of settlement, can be ruled out in a natural experiment where e.g. timing and relocation of immigrants are motivated by a policy free from economic considerations. The problems arising from the possible selection in the location of immigrants will be dealt with in Section 5.2.

3 History of Huguenot Migration to Prussia

This section summarizes the historical background of Huguenot immigration and provides narrative information of the consequences.

3.1 Immigration after the Edict of Potsdam

The persecution of Reformed Protestants in France started around 1530 and peaked at the St. Bartholomew's Day massacre of 1572 which was followed by a first wave of religious flight. From 1598 the Edict of Nantes granted religious freedom to the Huguenots until its revocation on October 18, 1685 by Louis XIV, the Sun King. Protestantism became illegal again and Huguenots were outlawed in the predominantly Catholic France. Protestant churches and schools were shut down and Huguenots once again became a target of persecution. While there was a constant outflow due to increasing harassment prior to the revocation, the movement grew into an exodus soon after. This was not anticipated by the King of France, who had assumed that only those people would leave who were in trouble with creditors or were without property and special skills, and therefore did not have strong connections to their homes (Scoville, 1960). Hence, he tried to force the Huguenots to convert to Catholicism. In spite of severe penalties such as life-long imprisonment, deportation into slavery or death, approximately 200,000 fled. Most of them settled in neighboring Protestant countries such as England, Germany, Ireland, the Netherlands and Switzerland.

The most famous example of those who offered refuge was Frederick William, the Great Elector of Brandenburg. Unlike his mostly Lutheran subjects, he was of Reformed faith and felt sympathy for his fellow Christians from France. Three weeks after Louis XIV revoked the Edict of Nantes, Frederick William issued the Edict of Potsdam offering his estates as a refuge to the Huguenots.¹

¹For a translated excerpt from the Edict see Appendix A

Of the estimated 43.000 Huguenots who left France² for the German territories, 16.000 to 20.000³ alone came to Brandenburg-Prussia, a country of approximately 1.5 million inhabitants at that time. Since there already were some French nobles living in Frederick William's court, Berlin became the final destination of many Huguenots, following the Edict of Potsdam. By the beginning of the eighteenth century, more than 5,000 Huguenots had settled in Berlin and its outskirts, making up to 20 percent of the town's total population. The rest settled in roughly 40 other towns and some few rural parishes. In total, about 90 percent of the Huguenots settled in towns.

Frederick William was anxious that the French would leave if they felt alienated by the natives. So he allowed them to build communities of refugees, so-called colonies, in each town of their settlement. These were parishes with their own church and service and, depending on the size, their own jurisdiction, police and schooling.

The literature agrees that the rich and powerful Huguenots mostly fled to England and the Netherlands. This picture is maintained by various descriptions of impoverished and half naked Huguenots arriving in Brandenburg, having lost everything during the flight. Nevertheless, Wilke (1988c) emphasizes that it was not only the poor nor the second-class nobility who came to Prussia. According to him, the Huguenots came to Prussia as a complete draft of society. He estimates that the immigrants were composed of 5% nobility, 7% mid-level functionary, 8% trade and manufacturing bourgeoisie, 20% workers and apprentices, 15% farmers and 45% small artisans and craftsmen in 1705.

These figures already draw a clear-cut picture of the occupational composition which resembled a town population much more than a rural society. There were two reasons for this: First of all, Huguenots were generally very well-educated and had adopted more skilled occupations in France.⁴ Second, in February 1686 Frederick William demanded

 $^{^{2}}$ Their origin was manifold; centers of emigration were the Languedoc (south), Dauphiné (south-east), the Champagne (north-east) and the Gascogne (south-west).

 $^{^{3}}$ Numbers vary with the inclusion of members of the military who were integrated into the Prussian army and thus not counted in colony lists.

⁴Scoville (1960) explains the economic advantage of the Calvinists over Catholics in France with their dominant role in public finance, their role as a "penalized minority", Protestant individualism and a Protestant ethic á la Max Weber. Incidentally, Scoville mentions Calvinist advocation of Bible reading. This might have translated into higher accumulation of human capital and skill (see Becker and Woessmann (2009) for similar arguments regarding Protestants in Prussia and Botticini and Eckstein (2005, 2007) for Jewish literacy and occupational selection).

that his delegates should refuse unskilled Huguenot workers to enter Brandenburg-Prussia (Mittenzwei, 1987).

3.2 The Economic Impact of French Immigrants

Frederick William, the Great Elector of Brandenburg, came into his reign in 1640 during the Thirty Years' War, which left the country depopulated and deserted after the Black Death had finally faded. The Margraviate of Brandenburg, Pomerania and Magdeburg, which made up most of his territory then, were hit hard by the war and suffered from the aftermath more than most other German states and kingdoms. Therefore, Frederick William and his successors became well-known for their repopulation policy (*Peuplierung*) and the intake of Huguenots was a major step to fulfil this aim.⁵

An increase in the population was perceived as a raise in the number of tax payers as well as a potential way to recruit more soldiers. Thus, the literature identifies economic motives for the intake of Huguenots (Jersch-Wenzel, 1978; Mittenzwei, 1987; Wilke, 1988a), while religious motives and sympathy towards fellow believers are not neglected. Skilled immigrants in particular were the most attractive targets and were expected to use their knowledge to set up and supervise manufactories. This was very much in line with the German economic thought of the seventeenth century (*Kameralismus*, a special kind of mercantilism) which was based on a positive balance of trade. The Huguenots were expected to produce 'domestic' goods that otherwise would have to be imported. Thus taking in the Huguenots, who were known to be good craftsmen, was an act of tolerance at first, but became an act of economic policy in hindsight.

Already in the Edict of Potsdam, Frederick William granted support and several privileges to all French refugees. This included exemptions from tariffs when entering the country, free use of abandoned houses and deserted land, exemption from all taxes and impositions except the consumption tax for 15 years, financial and material support for setting up businesses and manufactories, free land for those in agriculture and finally, freedom from guild coercion for 10 and later 15 years. All financial support was provided

 $^{{}^{5}}$ Frederick the Great pointed out at the beginning of his reign in 1740 that even after three regimes and nearly a century passed, the impact of the Thirty Years' War on the Margraviate, Pomerania and Magdeburg had not yet been made up for completely. Although massive efforts had been undertaken by each ruler to repopulate the land, it was not until the middle of the eighteenth century that the population reached pre-war levels (Franz, 1979, p. 100).

as a loan to be paid back once the businesses became profitable. This became necessary as many Huguenots had lost all of their possessions during the flight.⁶ Soon the Huguenots went into business and most of them resumed occupations they already had had in France - concentrating on textiles and apparel. Approximately 25.7 percent of the Huguenot craftsmen were occupied with the production of cloth and 32 percent with other textiles.⁷ As expected, the immigrants used their technological and managerial knowledge to set up manufactories, while the attempts made by their domestic counterparts were not able to surpass the lower stages of production (Jersch-Wenzel, 1978, p. 80).

Analyzing the economic impact of the Huguenots can only be attempted using historical sources. Unfortunately, most of the contemporary documentation seems to be somewhat clouded and biased in favor of self-adulation of the Prussian rulers. Consequently, the modern literature suffers from the lack of unbiased sources (Gwynn, 2001, p. 74).

This can be seen in some examples outlining the short-term benefits of hosting the Huguenots: When asked if his intentions to bring back Magdeburg to its former prosperity had been fulfilled, the King answered that the town had been idle for 40 years after the war, but when the refugees came all buildings had been filled within 18 years. Manufactories were established that had not been there before, foreign money had come to the town and hundreds of citizens were employed and contributed to consumption (Jersch-Wenzel, 1986, p. 163).

These statements are supported by a comparison of costs and benefits undertaken by the city council for the colony in Magdeburg in 1709, which found that Huguenot economic activities offset investments into them by far. In line with mercantile thinking, more people would lead to more wealth and costs caused by their privileges and subsidies should be offset by the additional consumption taxes. Based on the calculations of the city council, Jersch-Wenzel (1978) estimates an annual per-Huguenot return of 10 Thalers,

 $^{^{6}}$ Nevertheless Muret (1885) finds that some Huguenots purchased real estate, houses and manufactories with their own means and without subsidy.

 $^{^{7}}$ The data reflect the structure of Huguenot craftsmen in Berlin which, throughout the literature, is often used as a proxy representing all colonies in Brandenburg-Prussia. See Jersch-Wenzel (1978, pp. 72-74) for corresponding numbers in other professions.

throughout all colonies. This was approximately equal to the annual tax revenue obtained from every native. However, these calculations seem to be somewhat parsimonious and do not account for any external effects such as benefits resulting from technological diffusion.

The contemporary impressions of positive short-term benefits are nevertheless refuted rather than confirmed the in the modern literature. Refusals are mainly describe the ongoing attempts of Prussian rulers to hand out privileges and support to Huguenots to set up manufactories which seldom operated profitably and often went out of business soon after subsidies ran out (Jersch-Wenzel, 1978; Kindleberger, 1995; Scoville, 1960). The reasons for these failures were most often the lack of demand and markets for luxury goods, which were exactly the kind of products that were strongly supported by Prussian rulers. It was only the stocking production that succeeded in raising the necessary demand. Mittenzwei (1987, p. 124) suggests that Brandenburg-Prussia had not been ready for large-scale manufacturing at the beginning of the eighteenth century.

The long-term effects from immigration are similarly controversial. Mittenzwei (1987, p. 138) identifies four phases of Huguenot economic activity: A first phase of establishment from 1685 to the turn of the century, a boom phase in small-scale manufacturing up until 1735/36, a phase of decline up until 1767, and a phase of economic growth beyond the beginning of the nineteenth century. Mittenzwei's observation of growth around the turn of the nineteenth century is based on a massive increase in the number of looms for silk and cotton employed by members of the French colony in Berlin. She also observes a persistent downturn in the use of looms in the wool industry which was formerly dominated by Huguenots.

On the other hand, Jersch-Wenzel (1986) finds that the impact of the Huguenots on the Prussian economy and industry in particular lasted for nearly the whole eighteenth century but declined gradually towards the end. In 1797, a special commission filed a report stating that in the exact same way as the number of manufacturers had decreased in the colonies, the manufactories themselves were run-down (Jersch-Wenzel, 1986, p. 169). This impression might be due to increased assimilation. The homogeneity of the colony population eroded over time. Huguenots married into non-Huguenot families and left the community to live as normal Prussians and vice versa. Obviously, the manufactories had moved out of the colonies along with their entrepreneurs.

Though Jersch-Wenzel assumes that the commission had not overrated the declining impact of the Huguenot community, she suggests that the transfer of knowledge had a long-lasting impact. She concludes that knowledge and skill immigrated from France to Prussia and made a successful contribution to the Prussian economy. Wilke (1988a) confirms this by stating that the Huguenots brought the knowledge of production in centralized and decentralized manufactories to Prussia, a country that had not yet entered the stage of capitalist manufactories. Though they were not successful in establishing manufactories that endured over the long run (for aforementioned reasons), the Huguenots transferred their technological knowledge to their native apprentices and workers.

This very idea is also the target of our empirical research. We presume that even if any direct Huguenot influence on the economy vanished over time, their transferred knowledge was still active and had a positive impact on productivity in the manufacturing of textiles.

3.3 Knowledge Lead and Transfer

The diffusion of technical knowledge, once concentrated in France, is confirmed across all new host countries by Scoville (1952a). As for England, he notes that the Huguenots raised the quality of production and diffused skills that once were secrets of French manufacturers. In Holland, the silk and taffeta industry suddenly gained international reputation through Huguenot immigration. In Ireland, Huguenots gained massive influence on the manufacturing of linen and introduced new methods for spinning and weaving flax.

The economic situation in Ireland was most similar to Prussia at the time. Both suffered from the aftermath of a war, and just as for Ireland, it is generally agreed that Brandenburg-Prussia was a backward country at the end of the seventeenth century. Neither the putting-out system nor the cottage industry nor centralized manufacturing had advanced in Brandenburg-Prussia. In the late 1670s, Prussian functionaries built few manufactories in Berlin, otherwise there was no larger scale manufacturing. These state-forced enterprises were not driven by markets and thus either failed or performed dreadfully.

All in all, the Huguenots introduced both more advanced skills and new technologies. Bekmann (1751) found that they brought 46 professions to Brandenburg which were previously unknown to this country, most of them in the textile industries.⁸ One Huguenot carried with him the secret of dyeing fabrics in a special way, another brought the art of printing on cotton. Others introduced the hosiery knitting loom which replaced the manual production of stockings and socks. Furthermore, they introduced the knowledge of silk farming and silk spinning, a trade which was very important to Frederick William. He soon ordered the cultivation of mulberry trees in schoolyards to feed the silkworms and assigned special areas designated for plantation around Berlin.

While it seems to be agreed that the Huguenots were leaders in technical knowledge and skill in many trades, examples of actual transfers and diffusion of knowledge taking place are rare. The segregation of Huguenots into colonies might have imposed barriers to interactions with natives. Other obstacles to communication might have been the hostility displayed by Catholic and Lutheran natives, who would at times even refuse buy from the Reformed Huguenots.

Nevertheless, there are clear signs of frequent knowledge transfers between Huguenot artisans instructing native apprentices and workers. This form of interaction was strongly encouraged by Frederick William. When immigrants requested financial support to set up manufactories, many of the contracts included a fixed number of employees⁹ as well as the condition that they had to instruct native apprentices.¹⁰ In Halle on the Saale, it was publicly proclaimed that citizens should send their children to become apprentice to French manufacturers.

Even if these large-scale manufactories did not last for long, they resulted in the training of native apprentices and provision of new equipment. The equipment was

⁸Frederick the Great remarked that: "When Frederick William (the Great Elector) began his reign, this country was producing neither hats and stockings, nor serge or other woolen stuff; French diligence delivered all those goods to us. They fabricated cloth, screen cloth, serge, gentle cloth, drugs, griset, crepe, woven caps and stockings, beaver- and rabbit-hats, rabbit-hair hats and built dyeing works of all kind." Cited from Erbe (1937, p. 83).

⁹The entrepreneur Orelly was contracted to employ at least 8000 workers, André, Valentin and Claparède had to employ 110 looms.

 $^{^{10}}$ Mittenzwei (1987, p. 118) lists three examples of contracts including the order to employ native apprentices and to teach them the craft.

eventually sold or leased to either some native or otherwise mostly Huguenot craftsmen who set up smaller businesses which were far more successful.

As Scoville (1952a, p. 410) puts it, the rate of technological diffusion depended on the channels of communication between Huguenots and natives and on the size of the technological gap between France and the immigration country.¹¹ In the case of Prussia, the rate of diffusion was likely to be low. Direct communication between Huguenots and natives, other than the instruction of apprentices, was important to make the immigrants socially accepted and to raise demand for their products. Therefore, it was not until some years into the Huguenot refuge, when assimilation advanced and the native Prussians started to accept the French, that technological diffusion also progressed. Furthermore, the technological gap that separated Brandenburg-Prussia from France was large compared to other host countries like England and the Netherlands, and this state of underdevelopment prevented it from reaping immediate benefits from accommodating the Huguenots.

The technological change introduced by the Huguenots was likely to have been too abrupt to be applied in this country at once. This is in line with Becker et al. (2011) who find that the progress of the textile industry in Prussia was more incremental than disruptive. Nevertheless, the transfer of knowledge increased the rate of applied technological change and led to a higher growth equilibrium. As we will show subsequently, those towns with a higher share of first-generation Huguenot refugees became more productive than other towns in the long run.

3.4 The Settlement Pattern of Huguenots

The literature rarely touches on the question of why the Huguenots settled in certain towns. In general, rather than being able to select themselves into their settlement places, the Huguenots were channeled into those towns the Prussian rulers found most adequate. The Edict of Potsdam declared that the Huguenots were free to choose their place of settlement, but at the same time made recommendations for several towns¹²

¹¹Many others have found that the size of the technological gap determines the speed of a catch-up process (Gerschenkron, 1962; Findlay, 1978; Vandenbussche et al., 2006).

 $^{^{12}}$ See Appendix A for the corresponding paragraph in the Edict of Potsdam.

with sufficient livelihood (*Nahrung*).¹³ Many of the *bürgerliche Nahrungen* still remained vacant after the Thirty Years' War and the Huguenots were invited to fill these gaps.

Jersch-Wenzel (1978) assumes that the towns recommended in the Edict of Potsdam were chosen because they were the few bigger ones that could profit from the Huguenots. Klingebiel (1990) finds that the settlement pattern of the Huguenots reflected the structural requirements of the German regions after the Thirty Years' War. Schilling (1983, p. 9) identifies this as a case where an absolutist bureaucracy controlled the settlement of Huguenots and determined the scope and the direction of their economic activities.

To better understand how the immigration took place, we provide some examples from the contemporary literature. The Edict of Potsdam suggests that the flight to Brandenburg-Prussia was well organized by Frederick William. Already in the Edict, he advised the Huguenots from the north to head to Amsterdam where they would be welcomed by his delegates. From there they would be shipped through Hamburg into his realm. The Huguenots from the south were told to come to Frankfurt on the Main or Cologne where they would receive everything necessary and passage down the river Rhine to Cleves.¹⁴ The refugees usually moved in convoys from their home towns and arrived as groups at the assembly points where all immigrants were registered and their means and circumstances were recorded.

Afterwards, the Huguenots were assigned to a colony or settling place. According to Muret (1885), the welcoming delegates were to place the French where they would fit best and to transfer money required for their settlement from church collections. For example the commander of Lippstadt, Henri de Briquemault, placed all refugees from the Champagne region in the cities of Hamm, Soest, Minden and Lippstadt (Erbe, 1937, p. 34). As destination to host a large group of Huguenots which had to flee Mannheim¹⁵, the Great Elector suggested the cities of Prenzlau, Halle on the Saale, and Magdeburg. Two delegates visited these towns and decided that the entire colony from Mannheim would

 $^{^{13}}$ Nahrung was at this time defined as the occupation which one performed in order to subsist. When a village was granted market rights or town privileges, this was associated with the right to perform "bürgerliche Nahrungen" (crafts), as opposed to agriculture. However, the number of Nahrungen was limited to assure sufficient subsistence of the artisans and to guarantee the supply of the town population with the manufactured product for adequate prices. The supervising authority was usually the guild.

¹⁴For a more detailed description of migration routes see Klingebiel (2000).

¹⁵After their flight from France, a large group of Huguenots had settled in Mannheim (Palatine). When the French troops captured the town in 1689, the complete colony had to relocate again and decided to move to Brandenburg-Prussia.

move to Magdeburg (Gabriel, 1990), which had almost completely been destroyed in the war. As the Black Death had killed another 2650 inhabitants in 1683, the Huguenots were more than welcome. Another example is the rural French settlement in East Prussia. The Black Death was rampant there between 1708 and 1710 and depopulated a total of 8411 farms. Soon after Frederick I issued the call for new settlers, the Huguenots came and established themselves in the assigned areas of Insterburg and Gumbinnen.

The aforementioned facts lead to the conclusion that the place of settlement was not as arbitrary as announced in the Edict. The Huguenots were rather assigned to where they were needed most to repopulate and revitalize the deserted towns. These were exactly the towns depopulated by the Thirty Years' War and the Black Death. As repopulation was one of the crucial motives for attracting the Huguenots, they were clearly assigned to towns that had suffered the most losses. This settlement pattern will be useful for our identification strategy.

4 The Data

This section introduces two new data sources which we digitized for the purpose of this project.

In order to estimate the long-term effect of Huguenot immigration on productivity, we need data that measure productivity for a period that is often called the statistical dark age. We are able to calculate productivity at the firm level, using very early data on manufactories across Prussian towns in 1802.¹⁶ The manufacturing data are extracted from the "Register of Factories in the Prussian State" conducted by the Royal Prussian Secret Filing Department (Krug, 1805).¹⁷ To our knowledge, this is the earliest published overview of this kind in Prussia. The register includes all factories established within Prussian borders of 1802 except for those in Ansbach, Bayreuth, Neuchâtel, Silesia, and the new territories gained as compensation for losses in the war with France. ¹⁸

 $^{^{16}}$ 1802 was not a year with disturbing shocks to the Prussian economy. Those towns left of the river Rhine which had been annexed by Napoleon earlier are not included in the survey by definition. Other areas were not yet affected by the war and the Industrial Revolution had not started in Prussia.

¹⁷The department became the Prussian Statistical Office in 1805.

 $^{^{18}}$ We excluded 53, out of a total of 746 textile manufactories from the dataset in cases where manufactories were established in rural areas or in areas which did not belong to Prussia after 1807 (These are the spotted areas in Figure 2). We also excluded Huguenot settlements in rural areas since the occupation structure of rural colonies was very different.

During pre-industrial times, the expressions factory and manufactory were used synonymously in Prussia. However, there was a distinction between (manu)factory and craftsmen, where craftsmen produced on order and sold to a local demand, while (manu)factories produced larger quantities without order to satisfy national and even international markets (Hoffmann, 1969, p. 19). The latter form of production was also the criterion for inclusion in the survey.

The data includes the place and type of the manufactory, as well as the value of its manufactured goods, the value of raw materials used as inputs, the number of workers and the number of looms. Summary statistics are provided in Table 1. All manufactories were classified into 17 categories by their main input. Those manufactories classified as producing goods from wool, cotton, linen and silk represent our measure of textile manufacturing while all other categories will be used for a falsification test.

According to Krug (1805), the number of reported workers might be exposed to measurement error, mostly due to fluctuations during the year. For reasons unknown, data on the value of raw materials are missing for 96 of the textile manufactories. To be able to use a complete dataset, we impute missing values as described in Appendix B. We also have no information available on whether a manufactory was owned by or employed any Huguenots. However, as we aim to study the diffusion of technological knowledge it is not necessary to know about the physical presence of Huguenots in the production process

Data on the quantity of Huguenot immigration is very much unique for Brandenburg-Prussia. To our knowledge, none of the other host countries kept record of their immigrants. In Prussia, every French immigrant living in a colony was registered annually in the *Rôle général des Français refugiez dans les Estats de la Majesté le Roy de Prusse*. These immigration lists document the name of each Huguenot, the respective number of family members and servants, as well as his occupation. Because of continuous fluctuations in the first years (Jersch-Wenzel, 1985), we concentrate on the number of Huguenots living in towns and the number of Huguenots occupied in textiles in 1700 in order to estimate the impact of the first generation - the knowledge bearers.¹⁹

¹⁹The data source is Muret (1885).

To calculate the immigrant share at the town level, we use data from population censuses for Prussian towns provided in Schmoller (1922). Unfortunately, data on town population do not exist for 1700 and the first extensive census dates from 1730. The share of Huguenots in Prussian towns is thus defined as the number of Huguenots in 1700 over the town population in 1730. This definition would lead to an upward bias in the estimates only if the population in towns with Huguenot colonies systematically grew at a slower speed than others and vice versa.

In Figure 1 we present a map of Brandenburg-Prussia where grey areas depict her territory of 1685, the year the Edict of Potsdam was issued. Each town that subsequently hosted a Huguenot colony is marked with a cross. Most of the colonies that were founded after 1685 are located within these borders, except for the city of Stettin (Szczecin), whose colony was founded soon after the annexation of Swedish Pomerania in 1720. In Figure 2, towns with at least one textile manufactory are marked with a circle and towns with a Huguenot colony are again marked with a cross. We find that only eight towns hosting a Huguenot settlement did not subsequently develop large-scale textile manufacturing.

5 The Empirical Model

In this section we design the empirical model which will test our central hypothesis that a higher share of Huguenot population is associated with higher productivity in the long run.

5.1 The Productivity Model

We estimate productivity in textile manufacturing using a production function with technological progress:

$$\ln(\frac{Q}{L})_{ij} = \beta_1 \ln(A_j) + \beta_2 \ln(\frac{K}{L})_{ij}.$$
(1)

Productivity, defined as the ratio of output Q to labor L, is determined by the ratio of capital K to labor L as well as the level of technology A. Q is measured as the value of goods produced in manufactory i in town j. L is measured as the number of workers and K is represented by two variables: the number of looms and the value of materials used. Q, L and K are calculated in natural logarithms.

The share of Huguenots in a town's population enters as a measure determining the technology parameter A. The technological progress, brought by the Huguenots, came as a shock to the Prussian towns and varies with the ratio of Huguenots to natives, since technological diffusion is likely to increase with growing interaction possibilities. We assume that the exogenous technological progress caused by the Huguenots is Hicks-neutral. Both capital and labor are augmented; capital through the introduction of new and better looms and labor through the transfer of knowledge for more skillful application of the looms. This leads to the following estimating equation:

$$\ln(\frac{Output}{Worker})_{ij} = \beta_0 + \beta_1(\frac{Huguenots}{TownPopulation})_j + \beta_2 \ln(\frac{Materials}{Worker})_{ij} + \beta_3 \ln(\frac{Looms}{Worker})_{ij} + X'_j \gamma + u_{ij}.$$
(2)

X is a vector of characteristics that might have an influence on output and productivity (town size, availability of raw materials, religious composition of the population).

5.2 Exogenous Variation from Population Losses during the Thirty Years' War

In Chapter 3.4, we concluded that Huguenots, who came as an exogenous shock to the towns of Brandenburg-Prussia, were not able to select themselves into certain towns. However, if Prussian officials deliberately assigned Huguenots to towns for reasons that are unobserved but associated with productivity in textile manufacturing, estimation results might still be biased.

Even though their place of settlement was not randomly assigned, the Huguenots were channeled into towns which had been war-strapped and depopulated by plagues. We exploit this fact in an instrumental-variable strategy, where population losses during the Thirty Years' War serve as an instrument for the share of Huguenot population. This approach uses only that part of variation in Huguenot immigration that is due to the exogenous depopulation of a town during the war. The Thirty Years' War (1618-48) is a period of strong devastation and thus naturally one of the darkest spots in German demographic research. Even parish and tax registers, usually reliable sources for calculation of the population, are sparse. The only part of Prussia with sufficient information on population losses in towns is the Margraviate of Brandenburg. For other areas we have to draw on sources not exclusively taken for this purpose. We use population data for the closest pre-war date available and the closest post-war date available from the German handbook of towns (Keyser, 1939-1941), interpolate²⁰ them and calculate population losses at the town level. Where available, we also use data from Behre (1905) and Wohlfeil (1976) and calculate the average population loss over the three data sources (see Figure 3 for availability of the instrument).

The first stage of the instrumental-variable approach predicts the population share of Huguenots in a town j with the population decrease of the town during the war period:

$$\left(\frac{Huguenots}{TownPopulation}\right)_{j} = \pi_0 + \pi_1 PopLosses_j + X'_{ij}\delta + \nu_{ij}.$$
(3)

Exogeneity comes from the fact that the largest part of population losses did not emerge due to the act of war itself but through the occurrence of the Black Death in the 1620s and 1630s. Depopulation in this case can be interpreted as an event independent of a towns' economic activities and conditions, since epidemic mortality did not depend, for example, on social classes or size of the settlement (Voigtländer and Voth, 2009). The epidemics were spread by roaming troops, returning soldiers and fleeing peasants seeking refuge in towns (Pfister, 2007). The hygienic situation eventually translated into plague, dysentery and typhus and resulted in massive decimation. Moreover, high infant mortality reduced long-term population growth. Baptisms, as an indicator, remained very low even for the generation to follow. As we will try to demonstrate, our instrument is not affected by measures associated with textile production, and resulting estimates show the causal effect of Huguenot diffusion on textile manufacturing.

 $^{^{20}}$ See Appendix C for details. Interpolation increases the power of the instrument. However, using the original data does not affect the results substantially.

6 Results

This section provides the empirical analysis of the effects of Huguenot immigration on productivity. For this purpose, we exploit variation in Huguenot settlement and in the productivity of manufactories across Prussian towns between 1700 and 1802.

6.1 Basic Results

Table 2 shows basic results from OLS regressions for the 693 manufactories producing textiles across 302 Prussian towns. Throughout the regressions, we assume heteroskedasticity, since the errors might be correlated within towns. We therefore cluster standard errors at the town level.

A simple bivariate regression, shown in column 1, introduces the share of first generation Huguenots in the town's population in 1700. We find that the share of Huguenots is positively correlated with productivity in textile manufacturing in 1802.

When controlling for firm-level input factors in column 2, we find that the share of Huguenots remains significantly associated with productivity. The value of materials per worker adds a lot of power in predicting the productivity of a firm. The number of looms employed also has a positive but smaller effect. Manufactories that did not employ any looms appear to be even more productive. This is no surprise since these are only manufactories producing hats and gloves, and thus luxury goods with an high output value. Since we used imputational methods in cases where information on the value of input materials was missing, we include a dummy to identify these observations in each regression. This dummy is not significantly correlated with the dependent variable, which means that manufactories with missing input information did not systematically differ in their productivity.²¹

Our results indicate that an increase in the share of Huguenots by one percentage point translates into a 1.4 percentage point higher productivity in 1802, or alternatively, if Huguenot immigration increases by one standard deviation, productivity increases by 0.07 standard deviations.²² We interpret this result as a very conservative estimate

²¹Casewise deletion of observations with missing data (see Appendix B for further information) leads to similar results. The Huguenot coefficient is 1.405 with a t-statistic of 20.25.

 $^{^{22}}$ The mean share of Huguenots is 5.8 percent in a sample restricted to hosting towns.

because knowledge spill-overs to other towns might also have increased productivity there. It is therefore rather intriguing that productivity differences can still be observed after such a long time. Compared to an average annual European (agricultural) TFP growth of 0.2 percentage points (Persson, 2010), the gap does not seem too small.

Our estimates prove to be robust against the inclusion of several control variables.²³ In column 3, we control for the size of the town, since productivity and wages are usually higher in larger cities and a large town population might have an effect on prices of outputs sold and inputs purchased. Furthermore, input prices, especially in textiles, might be associated with availability of raw materials like wool. Thus, we include the number of sheep per capita at the county level. We find both variables not having an effect significantly different from zero. The size of the Huguenot coefficient is hardly affected.

The inclusion of the share of Protestants, which also might have had an effect on the diffusion of Huguenot knowledge since Protestants were probably less hostile than Catholics,²⁴ does not enter the model significantly (column 4). Furthermore, the inclusion of a dummy controlling for towns that did not belong to Prussia before 1700 does not change the coefficient for Huguenot diffusion. The intuition behind this dummy is to control for Prussian annexations after the big waves of Huguenot immigration. Very few colonies were established in these newly acquired towns. Additionally, this dummy controls for a possible east/west bias, since most of the territory acquired after 1700 is located in the former Kingdom of Poland.

If the settlement of Huguenots in Prussia only reflected the occurrence of pre-immigration textile production, our estimates would be driven by a path dependency prevalent in textiles. It might be that textile production in cities that hosted Huguenots achieved higher productivity in 1802 simply because they had been a center of textile production before their immigration. We deal with this concern by controlling for the progress attained in textiles before the Huguenots arrived. An Edict of 1680 documents the economic conditions of Brandenburg-Prussia, and found that due to the prevalent impact of the war on

 $^{^{23}}$ Unless specified otherwise, the source for the control variables is Mützell (1823-1825).

 $^{^{24}}$ Mokyr (1990) makes the point that Protestants are generally tolerant and thus more open to innovation and technological change.

most towns, the economy had still not reached the same level it had before 1618. The only craft of nationwide relevance was cloth production which was located in 24 towns (Mittenzwei, 1987). Since quantitative information of the state of textile production is not available for this time, we construct a dummy identifying those 24 towns. Column 5 in Table 2 shows the estimates when including the dummy. The dummy is not significantly associated with textile productivity in 1802, showing that pre-immigration textile towns are not the same as post-immigration textile towns.

In Table 3, we test whether the results hold when altering the explanatory variable of interest. As mentioned earlier, we use town population data from 1730 as the denominator, which might result in underestimating the effect if those towns hosting Huguenots systematically grew at a higher speed between 1700 and 1730. Using immigration list data from 1720 (GStA PK, 1720), we can calculate a more accurate ratio. Changing to the 1720 immigration list, we find the coefficient rising from 1.44 to 1.84, which hints at a downward bias in the results for 1700.

In column 2 we find that the share of Huguenots in 1795 does not have a significant effect on productivity, as opposed to earlier dates.²⁵ This means that the distant ancestors of the immigrants did not have a direct effect on productivity. There are several reasons for this: First of all, from 1720 on, newly immigrated Huguenots seem to have focused more on agriculture. Most of those immigrants were directed to rural settlements, but even for Berlin a shift in occupation towards farming could be observed. During the eighteenth century, the number of Huguenots working as farmers increased to 20 percent. These were mostly unskilled workers who were pushed off to the countryside (Wilke, 1988c, p. 58). Furthermore, the homogeneity of the group was eroded when natives married into the wealthy Huguenot families. From 1772 on, Prussians could become members of the colonies even if they were not Reformed Christians. Furthermore, many Huguenots left the colonies and became assimilated. Since only those refugees living in colonies were recorded on the lists, growing assimilation led to measurement error in the data. This might be the reason for the increased standard errors.

 $^{^{25}}$ The denominator here is town population in 1802.

While the share of Huguenots in a town's population is a good measure of immigration in general, such a variable neglects the possibility that only Huguenots employed in textiles might have transferred the relevant knowledge for textile production. We therefore use the number of Huguenots in textiles as an alternative variable of interest. The number of Huguenots employed in textiles in 1700 is positively associated with productivity in textiles in 1802 and the results shown in column 3 are qualitatively similar to previous estimations. Furthermore, we find that a dummy, reflecting Huguenot influence on a town in general achieves similar results (column 4). The variable of interest thus proves to be robust against changes in definition.

When running our different immigration measures against each other (column 5), we find that the effect is mainly absorbed by the immigrant share in 1700. We thus concentrate on the first generation of immigrants, who were the bearers of new technology and knowledge by definition.

6.2 Instrumental-Variable Results

So far, we included a measure of pre-immigration textile production in our estimations to control for the possibility that Huguenots were selected into towns with a prospering textile industry. Nevertheless, other possible patterns of selection might remain, and settlement could have been driven by unobserved factors that are not exogenous to productivity in 1802. We proceed using population losses during the Thirty Years' War as an instrumental variable.

As mentioned before, population data for the war period is very scarce and information is unavailable for many towns in our dataset, which results in a reduction of our sample size. Column 1 in Table 4 reports results of an OLS estimation when using the smaller sample for which data on population losses are available. Results in the small sample are similar to the large sample in Table 2; the coefficient of interest is slightly higher but the difference is not statistically significant. The reduced-form relationship between population losses in the Thirty Years' War and productivity in textile manufacturing 1802 (column 2) is positive and significant. The first stage of the instrumental-variable approach (column 3) shows that population losses can be used as an instrument for the share of the Huguenot population in 1700. A decrease in the population by 52 percentage points (the average losses across Prussian tonws) is associated with an increase of Huguenots in the town population by 5.3 percentage points. The second-stage estimate is significant and slightly higher than the OLS estimate.²⁶ We also report the Kleibergen-Paap test statistic for weak instruments in column 4. An F-statistic of 5.7 might raise concerns whether the instrument is sufficiently strong (Stock and Yogo, 2005).²⁷

Nevertheless, we can use the number of Huguenots occupied in textiles in 1700 as an alternative measure to the population share of Huguenots. In column 6, we find this measure of technological diffusion instrumented by the population losses during the Thirty Years' War being significantly associated with productivity. For this specification, the Kleibergen-Paap test has a value of 15.4 in the presence of clustered standard errors at the town level. Similar results are obtained when using a dummy for towns hosting a Huguenot colony (not shown).

6.3 Testing for Instrument Validity and further Discussion

This section provides a discussion and interpretation of the results and addresses some questions that might arise.

A falsification test, using non-textile manufactories might eliminate concerns of unobserved heterogeneity at the town level. If Huguenots settled in towns that subsequently established successful manufactories because of unobserved economic effects at the town level, we should also observe these effects on industries that were not advanced by immigration. Using information from non-textile manufactories that are also included in the 1802 survey, we show that the positive effect of Huguenot immigration was restricted to textile manufacturing. First, in Table 5 we show that Huguenot immigration had a significant positive effect on all subcategories of textile manufacturing. Second, in Table 6 we show that this positive effect cannot be observed in the non-textile sectors. Column

 $^{^{26}}$ To test if the results are driven by the massive Huguenot immigration and large-scale manufactories in Berlin, we also estimate the model excluding this city. All results remain qualitatively unaffected (not shown).

 $^{^{27}}$ When we employ robust standard errors instead of clustering at the town level, the model passes the weak instrument test using the proposed critical value of 10 (not shown).

1 shows estimates for all 695 non-textile manufactories. We do not find that Huguenot immigration had a significant effect on these manufactories. Further disaggregation into subcategories in columns 2-6 shows that the Huguenot effect cannot be observed in any of these industries.²⁸ Identification is thus unlikely to be driven by any unobserved town effects that affected all industries.

Throughout the literature we find examples of Huguenots who advanced other trades besides textiles in their host countries. In Brandenburg-Prussia, they were also known to be watchmakers, goldsmiths, wigmakers, tobacco farmers and producers of glass, paper, and small metal goods (needles and pins). Nevertheless, since we do not find any significant effect on non-textile industries, we conclude that the Huguenot immigration might not have been crucial for large-scale non-textile production.

As mentioned above, the reduced-form relationship between population losses and productivity is positive and significant. A resulting concern might be a violation of the exclusion restriction if the instrument had a direct effect on the outcome. If, for example, towns that suffered high population losses because of plagues subsequently experienced increasing real wages, decreasing interest rates or changes in the institutional framework and were thus able to become more productive, our estimates might reflect this effect (Pamuk, 2007). A positive selection might arise if Huguenots subsequently settled in these high-wage, low-interest towns. When excluding towns with Huguenot colonies from the regression, we find a negative but insignificant correlation between population losses during the Thirty Years' War and productivity in textile manufacturing (not shown). This means, if we only compare towns without Huguenot immigrations, we find that towns with higher population losses subsequently achieved (insignificant) lower productivity. We thus find no violation of the exclusion restriction.

If the instrument was affected by economic pre-conditions of a town that also affected productivity in textile manufacturing later on, the estimates might be biased. We try to test for this possibility by showing correlations between the instrument and pre-war conditions of the towns in Table 7.

 $^{^{28}}$ Nine other subcategories with very few observations are not shown, since sufficient asymptotic properties are missing to reasonably interpret the results. A positive correlation can be found in the paper and soap industries. Scoville (1952b) mentions that Huguenots in Brandenburg-Prussia produced soaps to wash and improve the quality of wool, so the soap industry might have been strongly affected by the textile industry.

Large towns might have a higher potential for population losses due to disease, but also for subsequent recovery. We show that the correlation between population losses and the size of the population before the outbreak of the war is negative and statistically insignificant at the town level (column 1).

Better accessibility of a town not only leads to more exchange of goods, it also leads to more exchange of germs and thus a higher risk of plague import. Our estimates might thus be biased if towns that were connected to roads lost a higher share of population during the war but also achieved higher productivity due to their integration in trade after the war. Using a seventeenth century map of Hanseatic trade routes (Bruns and Weczerka, 1962), we construct a variable that counts the number of roads entering a town. The town-level correlation between the number of roads and the population losses during the Thirty Years' War is positive but statistically insignificant (column 2).

Finally, a town that has existed for a long time might create institutions like markets, guilds and courts that lead to more interaction and thus a higher risk of diseases. We crudely measure institutions by the year when a town was first documented as holding city rights (Source: Keyser (1939-1941)). The town-level correlation between the documented year and population losses during the Thirty Years' War is small and statistically insignificant (column 3).

This means that it was neither the relatively bigger towns with a higher potential for recovery, nor those better connected in trade or with better institutions that suffered the most population losses during the war.

The dispersion of Huguenots all over Protestant Europe is likely to have led to the development of a transnational network of refugees. Ties with other Huguenot families might have led to advantages in international trade if immigrants were able to export their goods more successfully than their native counterparts. This means that even if immigration led to a long-term increase in productivity, it may not have been the result of technological diffusion but of network externalities.

If this was the case, the early Huguenot manufactories should have been most successful, since networks are likely to be the strongest between first generation immigrants. Many examples of failing manufactories during the early decades seem to rule out this possibility. It had been the lack of demand that drove almost all of these enterprises out of business. We support this argument by analyzing the year of establishment using a list of Prussian manufactories from 1769 (Hoffmann, 1969).²⁹ The average date of establishment of the 558 manufactories that provided this information is 1753 and only 8 were founded before 1700. If many of these manufactories were Huguenot owned, it was at least by the second generation.

In pre-industrial times, the transfer of tacit knowledge was often limited to cities or even city quarters and processes were kept as secret as possible, especially within guilds. Nevertheless, Huguenots probably traveled to other towns and diffused some of their knowledge there. Furthermore, natives that had acquired Huguenot knowledge and technology might have relocated. In any case, diffusion beyond town limits leads to an underestimation of the effect of Huguenot immigration.

7 Conclusion

The analysis undertaken in this paper empirically confirms the existence of positive longterm effects from skilled immigration on productivity. We show that textile manufactories in towns hosting a higher share of Huguenot-refugees in 1700 achieved higher levels of productivity in 1802. We interpret this result insofar that the immigration of highly skilled Huguenots led to technological diffusion and knowledge transfer between the Huguenots and the natives, which translated into a long-term increase in productivity of the textile sector. Most of the existing literature on this topic suffers from a possible bias in historical sources that drew a one-sided picture and concentrated on positive examples in order to please the King of Prussia - econometric evidence was missing completely.

We are able to connect town-level immigration data from 1700 to firm-level manufacturing data from 1802 to present a comparative analysis of the benefits from Huguenot migration over all Prussian colonies and towns. Our estimates suggest that there has indeed been a diffusion of technologies and knowledge resulting from the targeted immigration of skilled workers. Despite the possible technological diffusion to towns without Huguenot immigration, the impact of the knowledge transfer can still be observed in the

 $^{^{29}}$ Unlike the manufacturing data from 1802, the 1769 data do include the year of establishment.

original host towns more than a hundred years later. Moreover, the effect is restricted to the industry which was the main field of activity for the immigrants - textile manufacturing. These findings strongly support the assumption of intra-industry spill-overs from specialized immigrants.

Our results also contribute to the understanding of technological diffusion before the Industrial Revolution, when migration of journeymen and traveling apprentices was virtually the only way to diffuse new inventions and work processes. Unfortunately, these singular events are not documented and their consequences cannot be analyzed today. Therefore, the mass migration of Huguenots serves as a natural experiment that enables us to quantify the longevity of technological diffusion in pre-industrial times. While the attraction of journeymen and masters to prosperous towns results in highly endogenous knowledge transfers, we are able to identify variation in the immigration of skilled workers that is exogenous to the pre-migration success of a town.

Our empirical identification strategy employs the exogenous variation in Huguenot immigration that results from population losses due to Thirty Years' War and ensuing plagues, and allows us to interpret the results as a causal relationship. The relevance of our instrument is confirmed throughout the literature where the the immigration of Huguenots is often associated with the population losses during the war. Thus we are able to isolate a part of the variation in immigration that results from factors exogenous to the outcome - productivity in textile manufacturing.

The results confirm List's (1856) argument that Germany might owe some of her early growth to the immigration of skilled human capital, in a way that could not have been observed by contemporaries. This might be one of the rare examples in which we are able observe the transfer of knowledge through migration which was unaffected by any indirect means of communication. The effects of such transfers can be verified only in the long run and might often be neglected in short-term analysis.

References

- BECKER, S. O., E. HORNUNG, AND L. WOESSMANN (2011): "Education and Catch-up in the Industrial Revolution," *American Economic Journal: Macroeconomics*, 3, 92 126.
- BECKER, S. O. AND L. WOESSMANN (2009): "Was Weber Wrong? A Human Capital Theory of Protestant Economic History," *Quarterly Journal of Economics*, 124, 531 596.
- BEHRE, O. (1905): Geschichte der Statistik in Brandenburg-Preussen bis zur Gründung des Königlich Statistischen Bureaus, Carl Heymanns Verlag, Berlin.
- BEKMANN, J. C. (1751): Historische Beschreibung der Chur und Mark Brandenburg, vol. 1, Voß, Berlin.
- BEN-GAD, M. (2004): "The Economic Effects of Immigration A Dynamic Analysis," Journal of Economic Dynamics and Control, 28, 1825–1845.
- BODVARSSON, O. B. AND H. VAN DEN BERG (2009): The Economics of Immigration: Theory and Policy, Springer Verlag, Berlin and Heidelberg.
- BORJAS, G. J. (1994): "The Economics of Immigration," Journal of Economic Literature, 32, 1667–1717.
- (1995): "The Economic Benefits from Immigration," Journal of Economic Perspectives, 9, 3–22.
- (1999): "The Economic Analysis Of Immigration," in Handbook of Labor Economics, ed. by O. Ashenfelter and D. E. Card, Elsevier, vol. 3 A, 1697–1760.
- BOTTICINI, M. AND Z. ECKSTEIN (2005): "Jewish Occupational Selection: Education, Restrictions, or Minorities?" Journal of Economic History, 65, 922 – 948.
- (2007): "From Farmers to Merchants, Conversions and Diaspora: Human Capital and Jewish History." Journal of the European Economic Association, 5, 885 926.
- BRUNS, F. AND H. WECZERKA (1962): Hansische Handelsstrassen Atlas, vol. 1, Böhlau Verlag Köln Graz.
- CHELLARAJ, G., K. E. MASKUS, AND A. MATTOO (2008): "The Contribution of International Graduate Students to US Innovation," *Review of International Economics*, 16, 444 462.
- CHISWICK, C. U., B. R. CHISWICK, AND G. KARRAS (1992): "The Impact of Immigrants on the Macroeconomy," *Carnegie-Rochester Conference Series on Public Policy*, 37, 279–316.
- CIPOLLA, C. M. (1972): "The Diffusion of Innovations in Early Modern Europe," Comparative Studies in Society and History, 14, 46–52.
- CIRIACONO, S. (2005): "Migration, Minorities, and Technology Transfer in Early Modern Europe," Journal of European Economic History, 34, 43 – 64.
- DOLADO, J. J., A. GORIA, AND A. ICHINO (1994): "Immigration, Human Capital and Growth in the Host Country: Evidence from Pooled Country Data," *Journal of Population Economics*, 7, 193–215.
- DRINKWATER, S., P. LEVINE, E. LOTTI, AND J. PEARLMAN (2007): "The Immigration Surplus Revisited in a General Equilibrium Model with Endogenous Growth," *Journal of Regional Science*, 47, 569–601.
- EPSTEIN, S. R. (2004): "Property Rights to Technical Knowledge in Premodern Europe, 1300-1800," *The American Economic Review*, 94, 382–387.
- ERBE, H. (1937): Die Hugenotten in Deutschland, Essener Verlagsanstalt, Essen.
- FINDLAY, R. (1978): "Relative Backwardness, Direct Foreign Investment, and the Transfer of Technology: A Simple Dynamic Model," The Quarterly Journal of Economics, 92, 1–16.

- FOURIE, J. AND D. VON FINTEL (2011): "Settler Skills and Colonial Development," Working Paper 213, Economic Research Southern Africa.
- FRANZ, G. (1979): Der Dreißigjährige Krieg und das deutsche Volk, G. Fischer Verlag, Stuttgart, 4th ed.
- FRIEDBERG, R. M. AND J. HUNT (1995): "The Impact of Immigrants on Host Country Wages, Employment and Growth," *Journal of Economic Perspectives*, 9, 23–44.
- GABRIEL, M. (1990): "Die Hugenotten im Raum Magdeburg 1685-1985," in Die Hugenotten und das Refuge: Deutschland und Europa, ed. by F. Hartweg and S. Jersch-Wenzel, Colloquium Verlag, Berlin, 85–101.
- GERSCHENKRON, A. (1962): Economic Backwardness in Historical Perspective: A Book of Essays, Belknap Press, Cambridge.
- GSTA PK (1720): "Rôle général des Français refugiez dans les Estats de la Majesté le Roy de Prusse de L'annee 1720," I. HA Französisches Koloniedepartement, Rep. 122, Reg. 48, Nr. 1.
- GWYNN, R. (2001): Huguenot Heritage, Sussex Academic Press, Brighton, 2nd ed.
- HATTON, T. J. (2010): "The Cliometrics of International Migration: A Survey," Journal of Economic Surveys, 24, 941–969.
- HOFFMANN, H. (1969): Handwerk und Manufaktur in Preußen 1769, Akademie-Verlag, Berlin.
- HUNT, J. AND M. GAUTHIER-LOISELLE (2010): "How Much Does Immigration Boost Innovation?" American Economic Journal: Macroeconomics, 2, 31 – 56.
- JERSCH-WENZEL, S. (1978): Juden und "Franzosen" in der Wirtschaft des Raumes Berlin/Brandenburg zur Zeit des Merkantilismus, Colloquium Verlag, Berlin.
 - (1985): "Hugenotten, Juden und Böhmen in brandenburgischen Städten des 18. Jahrhunderts," in *Immigration et société urbaine en Europe occidentale XVIe-XXe siècle*, ed. by Étienne François, Recherche sur les Civilisations, Paris, 101–114.
 - (1986): "Ein importiertes Ersatzbürgertum? Die Bedeutung der Hugenotten für die Wirtschaft Brandenburg-Preußens," in *Die Hugenotten 1685-1985*, ed. by R. von Thadden and M. Magdelaine, Verlag Beck, München, 160–171.
- KEYSER, E. (1939-1941): Deutsches Städtebuch Handbuch städtischer Geschichte, Kohlhammer, Stuttgart.
- KINDLEBERGER, C. P. (1995): "Technological Diffusion: European Experience to 1850," Journal of Evolutionary Economics, 5, 229 242.
- KLINGEBIEL, T. (1990): "Aspekte zur Ansiedlung von Hugenotten in den norddeutschen Territorien," in Die Hugenotten und das Refuge: Deutschland und Europa, ed. by F. Hartweg and S. Jersch-Wenzel, Colloquium Verlag, Berlin, 67–79.
 - (2000): "Huguenot Settlements in Central Europe," in In Search of Peace and Prosperity: New German Settlements in Eighteenth-Century Europe and America, ed. by H. Lehmann, H. Wellenreuther, and R. Wilson, 39–67.
- KRUG, L. (1805): Betrachtungen über den Nationalreichtum des preußischen Staates und über den Wohlstand seiner Bewohner, vol. 2, J. F. Unger, Berlin.
- LIST, F. (1856): National System of Political Economy, J.B. Lippincott & Co.
- LITTLE, R. J. A. AND D. B. RUBIN (2002): *Statistical Analysis with Missing Data*, Wiley, New York, 2nd ed.

- MITTENZWEI, I. (1987): "Die Hugenotten in der gewerblichen Wirtschaft Brandenburg-Preußens," in Hugenotten in Brandenburg-Preußen, ed. by I. Mittenzwei, Akademie der Wissenschaften der DDR, Leipzig, 112–168.
- MOKYR, J. (1990): The Lever of Riches: Technological Creativity and Economic Progress, Oxford University Press.
- ——— (2002): The Gifts of Athena: Historical Origins of the Knowledge Economy, Princeton University Press.
- MÜTZELL, A. A. (1823-1825): Neues Topographisch-Statistisch-Geographisches Wörterbuch des Preussischen Staats, vol. 5 & 6, Karl August Kümmel, Halle.
- MURET, E. (1885): Geschiche der Französischen Kolonie in Brandenburg-Preußen, unter besonderer Berücksichtigung der Berliner Gemeinde, W. Bürgenstein, Berlin.
- NIEBUHR, A. (2010): "Migration and Innovation: Does Cultural Diversity Matter for Regional R&D Activity?" Papers in Regional Science, 89, 563–585.
- PAMUK, S. (2007): "The Black Death and the Origins of the 'Great Divergence' across Europe, 1300-1600," European Review of Economic History, 11, 289 317.
- PASERMAN, M. D. (2008): "Do High-Skill Immigrants Raise Productivity? Evidence from Israeli manufacturing Firms, 1990-1999," IZA Discussion Paper 3572, Boston University.
- PERI, G. (2011): "The Effect of Immigration on Productivity: Evidence from U.S. States," *Review of Economics and Statistics*, forthcoming.
- PERSSON, K. G. (2010): An Economic History of Europe: Knowledge, Institutions and Growth, 600 to the Present, Cambridge Univ. Press, Cambridge.
- PFISTER, C. (2007): Bevölkerungsgeschichte und historische Demographie 1500-1800, Oldenbourg, München.
- ROSENBERG, N. (1970): "Economic Development and the Transfer of Technology: Some Historical Perspectives," *Technology and Culture*, 11, 550–575.
- RUBIN, D. B. (1987): Multiple Imputation for Nonresponse in Surveys, Wiley, New York.
- SCHILLING, H. (1983): "Innovation through Migration: The Settlements of Calvinistic Netherlanders in Sixteenth- and Seventeenth-Century Central and Western Europe," *Histoire sociale-Social History*, 16, 7–33.
- SCHMOLLER, G. (1922): Deutsches Städtewesen in älterer Zeit, Kurt Schroeder / Bonn und Leipzig.
- SCOVILLE, W. C. (1951): "Spread of Techniques: Minority Migrations and the Diffusion of Technology," The Journal of Economic History, 11, 347–360.
- (1952a): "The Huguenots and the Diffusion of Technology. I," *The Journal of Political Economy*, 60, 294–311.
- (1952b): "The Huguenots and the Diffusion of Technology. II," *The Journal of Political Economy*, 60, 392–411.
- (1953): "The Huguenots in the French Economy, 1650-1750," The Quarterly Journal of Economics, 67, 423–444.
- (1960): The Persecution of Huguenots and French Economic Development 1680-1720, University of California Press.
- STATACORP (2009): Stata Multiple-Imputation Reference Manual Release 11., College Station, Texas.

- STOCK, J. H. AND M. YOGO (2005): "Testing for Weak Instruments in Linear IV Regression," in Identification and Inference for Econometric Models: Essays in Honor of Thomas Rothenberg, Cambridge Univ. Press, Cambridge, 80–108.
- VANDENBUSSCHE, J., P. AGHION, AND C. MEGHIR (2006): "Growth, Distance to Frontier and Composition of Human Capital," *Journal of Economic Growth*, 11, 97–127.
- VOIGTLÄNDER, N. AND H.-J. VOTH (2009): "The Three Horsemen of Growth: Plague, War and Urbanization in Early Modern Europe," CEPR Discussion Papers 7275, C.E.P.R.
- WILKE, J. (1988a): "Berlin zur Zeit des Edikts von Potsdam. Das Edikt und seine Bedeutung," in Hugenotten in Berlin, ed. by G. Bregulla, Union Verlag Berlin, 13–53.
- (1988b): "Der Einfluß der Hugenotten auf die gewerbliche Entwicklung," in *Hugenotten in Berlin*, ed. by G. Bregulla, Union Verlag Berlin, 227–280.
- (1988c): "Zur Geschichte der französischen Kolonie," in *Hugenotten in Berlin*, ed. by G. Bregulla, Union Verlag Berlin, 54–87.
- WOHLFEIL, R. (1976): "Kriegsverlauf 1635 bis 1642 Bevölkerungsverluste der brandenburgischen Städte zwischen 1625 und 1652/53. (Der Dreißigjährige Krieg II)," in Historischer Handatlas von Brandenburg und Berlin, ed. by H. Quirin, de Gruyter, Berlin, Lfg. 50.

Figure 1: Towns with Huguenot Colonies in Prussia, 1685-1795



The Map shows the territory of Brandenburg-Prussia at the time of the Edict of Potsdam in 1685. Urban Huguenot colonies that were founded after 1685 are marked by a cross. Source: Own illustration; see main text for details.





The Map shows the Prussian territory that was included in the survey in 1802. Spotted areas are excluded from our analysis. Towns with at least one textile manufactory are marked with a circle. Urban Huguenot colonies that were founded after 1685 are marked by a cross. Source: Own illustration; see main text for details.



Figure 3: Availability of Data on Population Losses during the Thirty Years' War'

The Map shows the Prussian territory that was included in the survey in 1802. Spotted areas are excluded from our analysis. Towns with at least one textile manufactory are marked with a circle. Towns for which population losses during the Thirty Years' War are known are marked with a plus. Source: Own illustration; see main text for details.

 Table 1: Descriptive Statistics

Variable	Mean	Std. Dev.	Min.	Max.	N
(ln) Output per worker	5.032	0.849	1.951	7.536	693
(ln) Value of materials per worker	4.470	0.947	0.887	7.271	693
(ln) Looms per worker	-1.146	1.296	-7.005	0.560	693
(ln) Workers	2.963	1.684	0	8.534	693
% Huguenots 1700	0.011	0.044	0	0.280	693
% Huguenots 1720	0.009	0.029	0	0.151	693
% Huguenots 1795	0.005	0.016	0	0.129	693
(ln) Huguenots in textiles 1700	0.425	1.248	0	6.047	693
(ln) Town population 1802	7.991	0.996	5.746	11.939	693
Merino sheep p.c. 1816 (county)	0.068	0.111	0	0.847	693
% Protestant	0.751	0.295	0.020	0.999	693
Not Prussia in 1720 (dummy)	0.348	0.477	0	1	693
Population losses in 30 Years' War	0.524	0.330	-0.591	0.925	186

Source: 1802 textile manufactories data taken from Krug (1805). Huguenot data taken from Muret (1885) and GStA PK (1720). All other data taken from Mützell (1823-1825), except for Population losses in 30 Years' War (see Appendix C for sources and construction of this variable). Missing data in the variable Value of materials per worker are imputed (see Appendix B for methodology). Output and Value of materials are measured in Prussian Thalers from 1802.

DepVar: (ln) Output per Worker	(1)	(2)	(3)	(4)	(5)
% Huguenots 1700	2.884^{***}	1.382^{***}	1.352^{***}	1.440^{***}	1.411***
	(0.666)	(0.088)	(0.156)	(0.171)	(0.206)
(ln) Value of materials per worker		0.800^{***}	0.800^{***}	0.801^{***}	0.800^{***}
		(0.021)	(0.022)	(0.022)	(0.022)
(ln) Looms per worker		0.062^{***}	0.063^{***}	0.062^{***}	0.062^{***}
		(0.014)	(0.015)	(0.015)	(0.015)
Not using looms (dummy)		0.231^{***}	0.233^{***}	0.235^{***}	0.235^{***}
		(0.039)	(0.040)	(0.039)	(0.040)
(\ln) Town population 1802			0.003	0.002	0.002
			(0.015)	(0.016)	(0.016)
Merino sheep p.c. 1816 (county)			0.093	0.103	0.104
			(0.186)	(0.191)	(0.191)
% Protestant				0.074	0.073
				(0.084)	(0.084)
Not Prussia in 1700 (dummy)				0.071	0.071
				(0.048)	(0.049)
Relevant textile production					0.010
before 1685 (dummy)					(0.048)
Dummy for imputed values		-0.002	-0.003	0.006	0.004
		(0.035)	(0.036)	(0.037)	(0.037)
Constant	5.001^{***}	1.442^{***}	1.413^{***}	1.328^{***}	1.333^{***}
	(0.041)	(0.110)	(0.143)	(0.136)	(0.138)
Observations	693	693	693	693	693
Number of towns	302	302	302	302	302
R-squared	0.02	0.84	0.84	0.85	0.85

Table 2: Huguenot Population Share and Productivity in Textile Manufactories in Prussia, 1802

Notes: Table shows OLS estimates at the firm level. Standard errors, clustered at the town level, in parentheses. Significance: *** p<0.01, ** p<0.05, * p<0.1. See main text for data sources and details.

DepVar: (ln) Output per Worker	(1)	(2)	(3)	(4)	(5)
% Huguenots 1700					1.851^{***}
% Huguenots 1720	1.839^{***}				(0.434) -1.226 (1.037)
% Huguenots 1795	(0.000)	1.109 (0.936)			-0.155 (0.990)
(\ln) Huguenots in textiles 1700		· · · ·	0.045**		-0.006
Huguenots 1700 dummy			(0.018)	0.135^{**} (0.055)	(0.022) 0.103^{*} (0.062)
Additional controls	yes	yes	yes	yes	yes
Observations	693	693	693	693	693
Number of towns	302	302	302	302	302
R-squared	0.84	0.84	0.84	0.84	0.85

Table 3: Alternative Measures of the Huguenot Share

Notes: Table shows OLS estimates at the firm level. Standard errors, clustered at the town level, in parentheses. Significance: *** p<0.01, ** p<0.05, * p<0.1. Additional controls: Value of materials per worker, looms per worker, dummy for not using any looms, town population 1802, sheep per capita, share of Protestants, dummy for towns not in Prussia 1720, dummy for towns with relevant textile production before 1685, a dummy for imputed values, and a constant. See main text for data sources and details.

	OLS		IV				
			% Huguenots	(ln) Output	(ln) Huguenots	(ln) Output	
DepVar:	(ln) Output	t per Worker	1700	per worker	in textiles 1700	per worker	
	Small Sample	Reduced Form	1st Stage	2nd stage	1st Stage	2nd stage	
	(1)	(2)	(3)	(4)	(5)	(6)	
% Huguenots 1700	1.582***			1.801**			
	(0.298)			(0.883)			
(\ln) Huguenots in textiles 1700	× /					0.078^{**}	
						(0.038)	
Population losses in 30 Years' War		0.182^{*}	0.101^{**}		2.321***	× ,	
		(0.098)	(0.042)		(0.592)		
(ln) Value of materials per worker	0.791^{***}	0.790^{***}	-0.001	0.791^{***}	0.075	0.784^{***}	
	(0.039)	(0.041)	(0.004)	(0.038)	(0.125)	(0.036)	
(ln) Looms per worker	0.109^{***}	0.127^{***}	0.012^{**}	0.106^{***}	0.111	0.119^{***}	
	(0.029)	(0.030)	(0.006)	(0.029)	(0.144)	(0.024)	
Not using looms (dummy)	0.359^{***}	0.390^{***}	0.019^{**}	0.355^{***}	0.398^{**}	0.359^{***}	
	(0.070)	(0.074)	(0.010)	(0.069)	(0.193)	(0.068)	
Relevant textile production	0.146^{*}	0.145^{*}	0.003	0.140^{*}	-0.368	0.174^{**}	
before 1685 (dummy)	(0.076)	(0.082)	(0.020)	(0.076)	(0.368)	(0.080)	
Additional controls	yes	yes	yes	yes	yes	yes	
Observations	186	186	186	186	186	186	
Number of towns	71	71	71	71	71	71	
R-squared	0.89	0.88	0.61	0.89	0.75	0.89	
Kleibergen-Paap F statistic				5.749		15.39	

Table 4: Instrumenting the Huguenot Population Share with Population Losses during the Thirty Years' War

Notes: Columns 1-2 show OLS estimates at the firm level. Columns 3-6 show the first and second stage estimates of an IV approach where population losses in the Thirty Years' War serve as an instrument. Sample: Towns with available data for the instrument. Standard errors, clustered at the town level, in parentheses. Significance: *** p < 0.01, ** p < 0.05, * p < 0.1. Additional controls: Town population 1802, sheep per capita, share of Protestants, dummy for towns not in Prussia 1720, a dummy for imputed values, and a constant. See main text for data sources and details.

	Wool	Linen	Cotton	Silk
DepVar: (ln) Output per Worker	(1)	(2)	(3)	(4)
% Huguenots 1700	0.706^{***}	3.102^{***}	1.967^{*}	1.699^{***}
	(0.243)	(0.503)	(1.035)	(0.549)
(ln) Value of materials per worker	0.818***	0.774***	0.744***	0.834***
	(0.028)	(0.038)	(0.143)	(0.024)
(ln) Looms per worker	0.086***	0.022	-0.011	0.495***
· / -	(0.017)	(0.079)	(0.128)	(0.066)
Not using looms (dummy)	0.257***	0.168	0.053	1.289***
	(0.040)	(0.362)	(0.746)	(0.171)
Additional controls	VOG	VOG	MOG	MOG
Additional controls	yes	yes	yes	yes
Observations	521	123	27	22
Number of towns	272	111	27	15
R-squared	0.84	0.88	0.89	0.99

Table 5: Huguenot Population Share and Productivity in Different Textile Manufactories

Notes: Table shows OLS estimates at the firm level. Standard errors, clustered at the town level, in parentheses. Significance: *** p<0.01, ** p<0.05, * p<0.1. Additional controls: Town population 1802, sheep per capita, share of Protestants, dummy for towns not in Prussia 1720, a dummy for imputed values, and a constant. See main text for data sources and details.

	(1)	(2)	(3)	(4)	(5)	(6)
DepVar: (ln) Output per Worker	Non-textile	Leather	Metal	Tobacco	Flour mills	Misc.
M						
% Huguenots 1700	0.415	-0.856	0.287	0.567	1.702	-0.013
	(0.336)	(0.597)	(0.562)	(0.529)	(1.437)	(0.697)
(ln) Value of materials per worker	0.813^{***}	0.826^{***}	0.822^{***}	0.849^{***}	0.880^{***}	0.592^{***}
	(0.019)	(0.023)	(0.039)	(0.081)	(0.060)	(0.109)
Additional controls	yes	yes	yes	yes	yes	yes
Observations	695	371	80	43	32	78
Number of towns	250	216	42	41	24	34
R-squared	0.89	0.89	0.95	0.90	0.97	0.72

Table 6: Huguenot Population Share and Productivity in Different Non-Textile Manufactories

Notes: Table shows OLS estimates at the firm level. Standard errors, clustered at the town level, in parentheses. Significance: *** p < 0.01, ** p < 0.05, * p < 0.1. Additional controls: Town population 1802, sheep per capita, share of Protestants, dummy for towns not in Prussia 1720, a dummy for imputed values, and a constant. See main text for data sources and details.

	Indicators of pre-War development					
	On Hanseatic Year city					
	(ln) City size	trade route	rights documented			
	(1)	(2)	(3)			
Population losses in 30 Years' War	-0.0061	0.0246	-0.0004			
	(0.0322)	(0.0169)	(0.0004)			
Observations	68	69	69			
R-squared	0.0005	0.031	0.013			

Table 7: Exogeneity of the Instrument

Notes: Table shows OLS estimates at the town level. Standard errors in parentheses. Constant not reported. Significance: *** p < 0.01, ** p < 0.05, * p < 0.1. See main text for data sources and details.

Appendix A Excerpt from the Edict of Potsdam

Article 3

German original

English translation

Weilen Unsere Lande nicht allein mit allen zu des Lebens Unterhalt erforderten Nothwendigkeiten wol und reichlich versehen, sondern auch zu établirung allerhand Manufacturen, Handel und Wandels zu Wasser und zu Lande sehr bequem, als stellen Wir denen, die darinn sich werden setzen wollen, allerdings frey, denjenigen Ort, welchen sie in Unserm Herzogthum Cleve, den Graffschafften Marck und Ravensberg, Fürstenthümern Halberstadt und Minden, oder auch in dem Herzogthum Magdeburg, Chur-Marck-Brandenburg und Herzogthümern Pommern und Preussen zu ihrer Profesion und Lebens Art am bequemsten finden werden, zu erwählen; Und gleichwie Wir dafür halten daß in gedachter Unserer Chur-Marck-Brandenburg die Städte Stendal, Werben, Rathenow, Brandenburg und Franckfurt und in dem Herzogthum Magdeburg die Städte Magdeburg, Halle und Calbe, wie auch in Preußen die Stadt Königsberg, so wol deßhalb weil daselbst sehr wolfeil zu leben als auch wegen der allda sich befindenden facilität zur Nahrung und Gewerb vor sie am bequemsten seyn werden Als haben Wir die Anstalt machen lassen befehlen auch hiemit und Krafft dieses so bald einige von erwehnten Evangelisch-Reformierten Französischen Leuten daselbst ankommen werden daß alßdan dieselben wol auffgenommen und zu allemdem so zu ihrem établissement nöthig ihnen aller Müglichkeit nach verholffen werden soll.

Because our country is convenient with everything one needs for a living and for establishment of manufactories, trade and commerce by water and land we make available for those who want to settle at whichever place they find in Our Duchy of Cleves, the Counties of Mark and Ravensberg, Principalities of Halberstadt and Minden or in the Duchy of Magdeburg, the Margraviate of Brandenburg and the Duchies of Pomerania and Prussia convenient for their profession and lifestyle; Although we recommend the cities Stendal, Werben, Rathenow, Brandenburg and Frankfurt in Our Margraviate of Brandenburg, Magedeburg, Halle and Calbe in the Duchy of Magdeburg, as well as the city of Königsberg in Prussia because they are most comfortable to live in as well as there is enough facility for food and craft and We already ordered and hereby command that as soon as some of the mentioned evangelic-reformed French people arrive, that they shall be accommodated and given everything needed and possible for their establishment (Own translation).

Appendix B Productivity Data and Imputation

The firm-level data used in this work was taken from the "Register of Factories in the Prussian State" conducted by the Prussian Royal Secret Filing Department in 1802 (Krug, 1805, pp. 219-381). The information was collected by inspectors who annually surveyed all manufactories in their area of responsibility and had to send in standardized and printed tables with the requested information on type, location, number of workers, number of looms, value of raw materials used, and value of production in Prussian Thalers. Manufactories are classified by their main input material into 19 categories (wool, linen, cotton, silk, leather, metals, oil-, groat-, and cutting mills, paper mills, tobacco, sugar, glass, soap, powder, earthenware, vinegar, wax, and miscellaneous). After we excluded manufactories in areas not belonging to Prussia after 1807 and in rural areas, textiles amount to 693 observations and non-textiles amount to 695 observations.

Historical records often suffer from missing data for reasons unknown and irreparable. The only category used in our empirical analysis which is missing data is the value of raw materials, where missings amount to 14 percent of observations. While the mechanism generating the missing data is unknown, we are able to observe a geographical pattern. Almost every province is missing few (2-5) observations, the exception being the provinces *Kurmark* and *Littauensches Department*, where all observations are missing. We assume that the assigned inspectors simply did not collect or report this information. This would imply that the values of the missing observations are not dependent on the value of the variable itself but on the location. Dropping all observations with missing data would reduce the sample size severely and introduce bias if the remaining observations are not representative of the full population of interest. This makes imputational methods the first choice to address the problem (Rubin, 1987; Little and Rubin, 2002).

We impute the missing data using univariate multiple imputation methods for continuous variables, integrated in Stata 11 (StataCorp, 2009). To attain a complete dataset, we impute missing data in the explanatory variable "Value of Materials" with all other variables used for the extensive regression in column 5 of Table 2, the value of outputs, the number of workers, the number of looms, the share of Huguenots, the town population, sheep per capita, the share of Protestants and a dummy for towns not belonging to Prussia before 1720.

Since the process that generated the missing values is unknown, the probability to have a missing value might depend on unobserved characteristics not included in the imputation. These unobservables again might influence output of the manufactory systematically. In such a case we would predict identical values for manufactories with identical observed but possibly different unobserved characteristics and bias the estimates in an unknown direction.

To make sure our estimates are not driven by imputed data, we include an imputation dummy in all of our regressions. The dummy becomes 1 if data on the value of raw materials used were originally missing for the observation.

Appendix C Population Loss Data and Construction of the Instrument

We compiled a database on population losses during the Thirty Years' War for those towns with textile manufactories in 1802. The data were assembled from three different sources, each providing a consistent overview over a certain area.

The most extensive source is the *Deutsches Städtebuch* (Handbook of German Towns) by Keyser (1939-1941). The compendium provides information for all German towns and includes data on population for various points in time. To calculate the population losses during the war period, we would ideally need population data for the years 1625 (Brandenburg did not enter the war until 1626) and 1652, which are the breaks also used

in other sources (Wohlfeil, 1976). Unfortunately, information about town population for the period in question is very rare and data for these exact years is even more scarce.

When available, we used data as close as possible to said years and interpolated them to match the beginning and the end of the war. The earliest date used is 1550 and the latest date is 1685. The interpolation was undertaken using population growth rates for Germany calculated in Pfister (2007).

Period	Growth Rate
1541 - 1550	7.2
1551 - 1560	7.1
1561 - 1570	5.8
1571 - 1580	4.6
1581 - 1590	4.1
1691 - 1600	3.2
1601 - 1625	3.2
1626 - 1650	-13.4
1651 - 1700	8-10

Table C-1: Population Growth in Germany (1914).

Growth rate in per mil calculated after Pfister (2007, p. 10)

For example, if a town had a population of 1000 in the year 1600, we use the growth rates to estimate a population of 1080 in 1625. If the population had reached 700 in 1660, we estimate a population of 650 in 1652. The population loss would thus be 40%, instead of 30% if we do not interpolate.

We only included towns if information on the number of residents, households, fireplaces or citizen was available both before and after the war. Finally we used only those pieces of information where the unit of observation was the same for both dates. Cases which, for example, reported the number of houses in a town before the war and the number of fireplaces after the war, were excluded. Comparability between towns with different units of observation is granted since we calculated growth rates. A total of 57 towns matched the criteria for inclusion. Interpolation increases the power of the instrument (F-test increases from 4.1 to 5.7). However, using the original data does not affect the results substantially.

The second source is a map by Wohlfeil (1976) showing the percentage of population losses in towns during the Thirty Years' War in the Margraviate of Brandenburg between 1625 and 1652/53. A total of 46 towns matched the criteria for inclusion.

The third source is a population table for towns in the *Kurmark* and the *Neumark* before and after the Thirty Years' War, published in Behre (1905). Here the number of residents is given for 1625 and 1645. A total of 37 towns matched the criteria for inclusion.

In total, we gathered information for a total of 71 different towns for which we also had data on textile manufactories and Huguenot immigration. If information for the same town was available from different sources, we calculated the mean to level possible measurement error.

Ifo Working Papers

- No. 113 Riener, G. and S. Wiederhold, Costs of Control in Groups, November 2011.
- No. 112 Schlotter, M., Age at Preschool Entrance and Noncognitive Skills before School An Instrumental Variable Approach, November 2011.
- No. 111 Grimme, C., S. Henzel and E. Wieland, Inflation Uncertainty Revisited: Do Different Measures Disagree?, October 2011.
- No. 110 Friedrich, S., Policy Persistence and Rent Extraction, October 2011.
- No. 109 Kipar, S., The Effect of Restrictive Bank Lending on Innovation: Evidence from a Financial Crisis, August 2011.
- No. 108 Felbermayr, G.J., M. Larch and W. Lechthaler, Endogenous Labor Market Institutions in an Open Economy, August 2011.
- No. 107 Piopiunik, M., Intergenerational Transmission of Education and Mediating Channels: Evidence from Compulsory Schooling Reforms in Germany, August 2011.
- No. 106 Schlotter, M., The Effect of Preschool Attendance on Secondary School Track Choice in Germany, July 2011.
- No. 105 Sinn, H.-W. und T. Wollmershäuser, Target-Kredite, Leistungsbilanzsalden und Kapitalverkehr: Der Rettungsschirm der EZB, Juni 2011.
- No. 104 Czernich, N., Broadband Internet and Political Participation: Evidence for Germany, June 2011.
- No. 103 Aichele, R. and G. Felbermayr, Kyoto and the Carbon Footprint of Nations, June 2011.
- No. 102 Aichele, R. and G. Felbermayr, What a Difference Kyoto Made: Evidence from Instrumental Variables Estimation, June 2011.
- No. 101 Arent, S. and W. Nagl, Unemployment Benefit and Wages: The Impact of the Labor Market Reform in Germany on (Reservation) Wages, June 2011.

- No. 100 Arent, S. and W. Nagl, The Price of Security: On the Causality and Impact of Lay-off Risks on Wages, May 2011.
- No. 99 Rave, T. and F. Goetzke, Climate-friendly Technologies in the Mobile Air-conditioning Sector: A Patent Citation Analysis, April 2011.
- No. 98 Jeßberger, C., Multilateral Environmental Agreements up to 2050: Are They Sustainable Enough?, February 2011.
- No. 97 Rave, T., F. Goetzke and M. Larch, The Determinants of Environmental Innovations and Patenting: Germany Reconsidered, February 2011.
- No. 96 Seiler, C. and K. Wohlrabe, Ranking Economists and Economic Institutions Using RePEc: Some Remarks, January 2011.
- No. 95 Itkonen, J.V.A., Internal Validity of Estimating the Carbon Kuznets Curve by Controlling for Energy Use, December 2010.
- No. 94 Jeßberger, C., M. Sindram and M. Zimmer, Global Warming Induced Water-Cycle Changes and Industrial Production – A Scenario Analysis for the Upper Danube River Basin, November 2010.
- No. 93 Seiler, C., Dynamic Modelling of Nonresponse in Business Surveys, November 2010.
- No. 92 Hener, T., Do Couples Bargain over Fertility? Evidence Based on Child Preference Data, September 2010.
- No. 91 Schlotter, M. und L. Wößmann, Frühkindliche Bildung und spätere kognitive und nichtkognitive Fähigkeiten: Deutsche und internationale Evidenz, August 2010.
- No. 90 Geis, W., High Unemployment in Germany: Why do Foreigners Suffer Most?, August 2010.
- No. 89 Strobel, T., The Economic Impact of Capital-Skill Complementarities in German and US Industries Productivity Growth and the New Economy, July 2010.
- No. 88 Falck, O., M. Fritsch and S. Heblich, The Phantom of the Opera: Cultural Amenities, Human Capital, and Regional Economic Growth, June 2010.